

City Research Online

City, University of London Institutional Repository

Citation: O'Logbon, J., Newlove-Delgado, T., McManus, S., Mathews, F., Hill, S., Sadler, K. & Ford, T. (2022). How does the increase in eating difficulties according to the Development and Well-Being Assessment screening items relate to the population prevalence of eating disorders? An analysis of the 2017 Mental Health in Children and Young People survey. International Journal of Eating Disorders, 55(12), pp. 1777-1787. doi: 10.1002/eat.23833

This is the published version of the paper.

This version of the publication may differ from the final published version.

Permanent repository link: https://openaccess.city.ac.uk/id/eprint/29150/

Link to published version: https://doi.org/10.1002/eat.23833

Copyright: City Research Online aims to make research outputs of City, University of London available to a wider audience. Copyright and Moral Rights remain with the author(s) and/or copyright holders. URLs from City Research Online may be freely distributed and linked to.

Reuse: Copies of full items can be used for personal research or study, educational, or not-for-profit purposes without prior permission or charge. Provided that the authors, title and full bibliographic details are credited, a hyperlink and/or URL is given for the original metadata page and the content is not changed in any way.
 City Research Online:
 http://openaccess.city.ac.uk/
 publications@city.ac.uk

Revised: 5 October 2022

108x, 0, Downloaded from https:/

ORIGINAL ARTICLE

EATING DISORDERS WILEY

How does the increase in eating difficulties according to the Development and Well-Being Assessment screening items relate to the population prevalence of eating disorders? An analysis of the 2017 Mental Health in Children and Young People survey

Jessica O'Logbon MPhil¹ | Tamsin Newlove-Delgado PhD, MPH, MBChB² | Sally McManus MSc³ | Frances Mathews MRes, BSc (Hons)² | Suzanne Hill PhD, MSc, Bsc³ | Katharine Sadler MSc, Bsc³ | Tamsin Ford PhD, FRCPsych, FMedSci¹

¹Department of Psychiatry, University of Cambridge, Cambridge, UK

²Medical School, University of Exeter, Exeter, UK

³National Centre for Social Research, London, UK

Correspondence

Jessica O'Logbon, Department of Psychiatry, University of Cambridge, Herchel Smith Building for Brain & Mind Sciences, Robinson Way, Cambridge CB2 0SZ, UK. Email: jo458@cantab.ac.uk

Funding information

Department of Health and Social Care; Medical Research Council, Grant/Award Number: MR/ V027751/1; NIHR Advanced Fellowship, Grant/Award Number: NIHR300056; NIHR Cambridge Biomedical Research Centre, Grant/Award Number: BRC-1215-20014

Action Editor: Ruth Striegel Weissman

Abstract

Objective: We examine the test accuracy of the Development and Well-Being Assessment (DAWBA) eating disorder screening items to explore whether the increased eating difficulties detected in the English National Mental Health of Children and Young People (MHCYP) Surveys 2021 reflect an increased population prevalence.

Methods: Study 1 calculated sensitivity, specificity, and positive and negative predictive values from responses to the DAWBA screening items from 4057 11–19-year-olds and their parents, in the 2017 MHCYP survey. Study 2 applied the positive predictive value to data from 1844 11–19-year-olds responding to the 2021 follow-up to estimate the prevalence of eating disorders in England compared to 2017 prevalence.

Results: Parental report most accurately predicted an eating disorder (93.6%, 95% confidence interval: 92.7–94.5). Sensitivity increased when parent and child answers were combined, and with a higher threshold (of two) for children. The prevalence of eating disorders in 2021 was 1% in 17–19-year-olds, and .6% in 11–16-year-olds—similar to the prevalence reported in 2017 (.8% and .6%, respectively). However, estimates for boys (.2%–.4%) and young men (.0%–.4%) increased.

Discussion: We found tentative evidence of increased population prevalence of eating disorders, particularly among young men. Despite this, the DAWBA screening items are useful for ruling *out* eating disorders, particularly when parents or carers screen negative, but are relatively poor at predicting who will have a disorder. Data from both parents and children and applying a higher cut point improves accuracy but at the expense of more missed cases.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2022 The Authors. International Journal of Eating Disorders published by Wiley Periodicals LLC.

Public Significance Statement: The prevalence of eating disorders did not markedly change from 2017 to 2021, but we found tentative evidence of an increase, particularly among young men. This is despite larger increases in problematic eating, which need further investigation. The DAWBA screen is best suited to ruling out eating disorders which limits its clinical applications as it would provide many false positives requiring further assessment.

KEYWORDS

eating behavior, eating disorder, prevalence, screening, survey

1 | INTRODUCTION

The prevalence of eating disorders peaks during adolescence (Potterton et al., 2020)-a life stage associated with important milestones and transitions, and during which impaired functioning can catastrophically undermine subsequent health, educational, and social outcomes. Eating disorders are pernicious, multi-factorial conditions whose complexity is further compounded by their high rate of comorbidity with other psychiatric disorders, especially depression and anxiety (Mitchell et al., 2014). Given the resultant high morbidity and mortality (Fisher et al., 2001; Petkova et al., 2019), early identification and prompt treatment are crucial. Experts have recently suggested that we have greatly underestimated the prevalence of eating disorders, due to stigma and the tendency to conceal difficulties (Zipfel et al., 2022). Worryingly, presentations of young people with eating disorders to health services increased rapidly during the Coronavirus (COVID-19) pandemic in both high- and low-income countries (Feinmann, 2021). For example, reports indicate a doubling in admissions for re-feeding in Australia (Haripersad et al., 2021) and urgent referrals in England during 2020, combined with a smaller increase in nonurgent referrals (NHS England, 2022). While the number of young people seeking treatment has increased, only assessments of population-based samples can differentiate between an increase in the underlying symptomology or a change in treatmentseeking behavior, which is important to clarify so that policy and commissioning responses are evidence based.

The Mental Health of Children and Young People in England (MHCYP) survey (Vizard et al., 2018) was commissioned to estimate the prevalence of mental health conditions, including eating disorders. The MHCYP surveys used the Development and Well-Being Assessment (DAWBA), a multi-informant standardized diagnostic assessment, to assess mental health among a probability sample of 9117 2-to-19-year-olds. The DAWBA (Goodman et al., 2000) includes structured and semi-structured questions within modules that cover most mental health conditions. There are parallel versions for children and young people aged 11 years or more and parents/carers, which can be administered via interview or completed online. A brief questionnaire version is available for teachers. Each module includes "screening items," which aim to select those reporting any difficulties related to that disorder for further detailed structured questions and semi-structured probes. Clinical raters assess data from all informants to make a clinical judgment about the likelihood that the child or young person meets diagnostic

criteria for that disorder. The DAWBA has been widely used in clinical practice and research (Aebi et al., 2012; Moya et al., 2005).

Given the international rise in presentations to services with eating disorders during the COVID-19 pandemic (Feinmann, 2021; Zipfel et al., 2022), the follow ups of MHCYP in 2021 and 2022 included the DAWBA eating disorder screening items to allow direct comparison of eating difficulties with 2017. Unfortunately, we lacked time or funding to complete the full DAWBA eating disorder module for those who screened positive, but policymakers and commissioners need to understand how screening positive relates to the prevalence of eating disorders in this population.

The 2021 follow-up survey report has indicated a doubling in the proportion of 11–16-year-olds screening positive on the DAWBA eating disorder screen since 2017 (from 6.7% to 13.0%) and an increase from 44.6% to 58.2% among 17–19-year-olds (Williams et al., 2021). It is crucial that we understand how these reports of eating difficulties on the screening items predict an eating disorder diagnosis.

We examined the diagnostic accuracy of the DAWBA screening questions, with the aim of estimating the prevalence of eating disorders from the 2021 follow-up data. Our objective was to provide empirical context to better assess the extent to which increased clinical demand is being driven by increased population prevalence and to examine the impact of COVID-19 (Newlove-Delgado et al., 2021; Williams et al., 2021), by comparing the estimated prevalence to that measured in 2017. Study 1 explored whether the informant (parent or child/young person), the threshold score, or the diagnostic classification influenced the diagnostic accuracy of the DAWBA eating disorders screen and addressed this by using two different false negative rates (described in *Methods*). Study 2 then applied the Positive Predictive Values (PPV) generated from Study 1 to estimate the prevalence of eating disorders in 2021 compared to that in 2017.

We hypothesized that:

- 1. Parent and child reports combined would yield the highest diagnostic accuracy.
- The standard DAWBA screen threshold of one positive answer for children/young people would capture more cases than a cut point of two (currently used for parents).
- Diagnostic accuracy would not differ between the Diagnostic and Statistical Manual of Mental Disorders, 5th edition (DSM-5)

(American Psychiatric Association, 2013) or the International Classification of Diseases, 10th edition (ICD-10) (World Health Organization, 1992).

4. Estimates of eating disorder prevalence would suggest a significant population increase.

2 | METHODS

2.1 | Ethical considerations

The survey was reviewed and approved by the West London & GTAC Research Ethics Committee (16/LO/0155) and the Health Research Authority Confidentiality Advisory Group (16/CAG/0016). The follow-up surveys were approved by the Office for National Statistics Ethical Committee, and our analysis by the University of Cambridge Psychology Research Ethics Committee (PRE.2020.145).

2.2 | Study 1: The accuracy of the DAWBA screen in MHCYP 2017

The 2017 MHCYP survey involved a stratified probability sample of 9117 children and young people aged 2–19 years living in England drawn from the NHS Patient Register. More detail about the sampling approach and sample are available in the NHS Digital report (Vizard et al., 2018). Children or young people, and one of their respective parents were invited to complete the DAWBA interview face-to-face with trained lay interviewers. For children who were aged 16 and under, parents were interviewed first with permission sought from the parent to interview their child. Children provided assent. Conversely, 17–19-year-olds were directly asked for their consent, with permission subsequently sought for their parents to be interviewed. We report results for the 11–16-year-olds and the 17–19-year-olds separately due to this difference in assessment.

Each module of the DAWBA (including behavioral disorders, anxiety, depression, and neurodevelopmental disorders as well as eating

	Question	Ans	wer
		No	Ye
1.	Have you ever thought you were fat even when other people told you that you were <i>very</i> thin?	0	1
2.	Would you be ashamed if other people knew how much you eat?	0	1
3.	Have you deliberately made yourself vomit (throw up)?	0	1
4.	Do your worries about eating (What? Where? How much?) really interfere with your life?	0	1
5.	If you eat too much, do you blame yourself a lot?	0	1

For parents: if two or more of the questions is answered 'Yes', continue. Otherwise skip to next section.

N.B. Questions for parents used 'your child' or the child's name

FIGURE 1 The Development and Well-Being Assessment eating disorder screening questions 3

4 WILEY-EATING DISORDERS

disorders) applies questions to identify those with no problems to proceed directly to the next module, reducing participant burden. Those who screened positive are asked more detailed structured questions that relate directly to the diagnostic criteria, while semi-structured probes explore the experience of these difficulties (see full DAWBA eating disorders module in Data S1). A small team of expert clinical raters drawn from psychiatry, pediatrics, and public health (including TF and TND) reviewed data from informants to assign diagnoses according to DSM-5 and ICD-10 criteria, which slightly differ (see Data S2).

The DAWBA eating disorder module was developed for the 2004 British Child and Adolescent Mental Health Survey (BCAMHS), which involved data from a community sample of 500 young people and their parents (Meltzer et al., 2003), and a sample of 174 Brazilian girls aged 7–17 (48 with eating disorders, 55 clinical controls, and 71 community controls; Moya et al., 2005). This work established that a threshold of two positive answers for parents or one for children/ young people, for the five screening questions (see Figure 1), combined with the rest of the eating disorders module, resulted in specificity and sensitivity for a diagnosis of any eating disorder of 94% and 100%, respectively (Moya et al., 2005). The rationale behind the lower threshold for young people is based on the complex, often wellhidden symptomology of eating disorders (Couturier & Lock, 2006; Vandereycken & van Humbeeck, 2008; Viglione et al., 2006).

2.3 | Diagnostic accuracy of the eating disorder screening items in the MHCYP 2017

We use the clinically rated multi-informant DAWBA diagnoses of eating disorders in MHCYP 2017 as the reference standard. We provide diagnostic accuracy measures based on children who were diagnosed with any eating disorder included in the ICD-10 or DSM-5. There were very few eating disorders cases as this was an epidemiological sample, so sub-type analyses may have revealed the identity of the patients.

The index test was whether the informant scored above or below the threshold on the DAWBA screening questions: (i) as it is normally applied or (ii) set at two positive items for both parents and young people. We assessed the diagnostic accuracy of (1) these two different thresholds, (2) ICD-10 and DSM-5 criteria, and (3) individual responses versus combined responses from the parent and child or young person (where available). The components of diagnostic accuracy are defined in Figure 2, along with how they are calculated. They include PPV, Negative Predictive Value (NPV), specificity and sensitivity. Sensitivity is the ability of a test to correctly identify patients with a condition (percentage of true positives) while specificity is the ability to correctly identify people without the condition (percentage of true negatives). PPV and NPV are the probabilities that the test correctly identifies an individual as having a specific condition or not following a positive or negative test result, respectively. The overall diagnostic accuracy of the DAWBA screening questions was its ability to detect a condition when it is present and the absence of a condition when it is absent.

The 2017 MHCYP routed all children and parents who screened negative through to the next module, which led to zero false negatives. A study designed to establish test accuracy would collect data on some screen negatives but because this was a survey that was not designed to do that, we applied false negative rates from a previous validation study of the DAWBA screen based on the 2004 BCAMHS (Meltzer et al., 2003). This involved 500 participants from a community-based sample and 41 participants from a clinical sample who completed the full DAWBA eating disorders module with no skip rules (Meltzer et al., 2003). Applying the screen to these data produced no false negatives in the community sample but failed to detect 1/41 (2.4%) eating disorders in the clinical sample. Therefore, we ran our analysis twice, once assuming zero false negatives (0%) and once assuming a false negative rate of 2.4%.

	Has an eating disorder	Does not have an eating disorder	
	uistiuci		1
Screened positive	True positive (A)	False positive (B)	= Positive predictive value
on the DAWBA			$\frac{A}{(A+B)} X 100$
Screened negative	False negative (C)	True negative (D)	= Negative predictive value
on the DAWDA			$\frac{D}{(C+D)} X 100$
	$= \text{Sensitivity}$ $\frac{A}{(A+C)} X \text{ 100}$	$= Specificity$ $\frac{D}{(B+D)} X 100$	Overall diagnostic accuracy $\frac{A/C}{B/D}$





FIGURE 3 Flow diagram and Development and Well-Being Assessment administration for each study

TABLE 1 Measures of diagnostic accuracy of the standard DAWBA score threshold for an eating disorder diagnosis according to DSM-5 and ICD-10 criteria for children aged 11–16 years old, young people aged 17–19 years old and their respective parents (with 95% confidence intervals) in 2017 based on zero false negatives

	Diagnostic criteria used for clinical	Positive predictive	Negative predictive			
Informant type	rating	value (%)	value (%)	Sensitivity (%)	Specificity (%)	Accuracy (%)
Parents of 11–16-year-	DSM-5	4.9 (1.9–7.9)	99.8 (99.6–100.0)	66.7 (42.8-90.5)	93.7 (92.8–94.6)	93.6 (92.7-94.5)
olds (n = 3112)	ICD-10	4.4 (1.6-7.2)	99.8 (99.6-100.0)	64.3 (39.2-89.4)	93.7 (92.8-94.6)	93.6 (92.7-94.5)
Children aged 11-	DSM-5	1.0 (.9–1.1)	100.0 (-)	100.0 (66.4-100.0)	65.7 (63.4-67.5)	65.8 (63.9-67.6)
16 years ($n = 2597$)	ICD-10	.9 (-)	100.0 (-)	100.0 (63.1-100.0)	65.6 (63.8-67.5)	65.7 (63.9-67.6)
Parents and their	DSM-5	.8 (.7–.8)	100.0 (-)	100.0 (66.4–100.0)	55.2 (53.3-57.1)	55.4 (53.4-57.3)
children 11–16 years combined ($n = 2591$)	ICD-10	.7 (.6–.7)	100.0 (-)	100.0 (63.1-100.0)	55.2 (53.2-57.1)	55.3 (53.4-57.2)
Parents of 17–19-year- olds ($n = 415$)	DSM-5 & ICD-10 ^a	6.3 (4.5-8.6)	100.0 (-)	100.0 (15.8–100.0)	92.7 (89.8-95.1)	92.8 (89.8–95.1)
Young people aged 17–19 years old (n = 935)	DSM-5 & ICD-10 ^a	1.7 (1.6–1.8)	100.0 (-)	100.0 (59.0-100.0)	55.3 (52.0-58.5)	55.6 (52.4-58.8)
Parents and their children aged 17–19 years combined (n = 413)	DSM-5 & ICD-10 ^a	.9 (.8-1.0)	100.0 (-)	100.0 (15.8–100.0)	47.9 (43.0-52.9)	48.2 (43.3-53.1)

^aBoth diagnostic criteria detected the same cases.

5

(098108x, 0, Downloaded from https://onlinelibiary.wiley.com/doi/10.1002/eat.23833 by City, University Of London, Wiley Online Library on [02/11/2022]. See the Terms and Conditions

(https://or

ibrary.wiley.com

) on Wiley Online Library for rules

of use; OA articles are governed by the applicable Creative Comm

6 WILEY-EATING DISORDERS

2.4 | Study 2: Application of estimated PPVs to the 2021 MHCYP survey findings

All participants who consented to re-contact in 2017 were invited by mail to complete a brief questionnaire, which included the eating disorders screening questions. As previously, parents reported for their children under the age of 16 and young people aged 11 and over completed their own reports. In 2021, a total of 3667 participants completed the follow-up survey, which included 1844 participants aged 11–19 (Williams et al., 2021). Figure 3 shows the flow diagrams of participants and methods for both Studies 1 and 2.

Since none of the participants in 2021 could be diagnosed with an eating disorder, we applied the estimated PPVs from Study 1 to those who "screened positive" in this age group in MHYCP 2021 to provide a crude estimate of the proportion of those screening positive in 2021 whom we would expect to meet diagnostic criteria for an eating disorder if they had completed the full DAWBA eating disorders module. We used the PPVs from parent-reported estimates for 11-16-year-olds, and self-reported estimates for 17-19-year-olds, as data for only these groups were available in the published 2021 report (raw data not yet available for analysis) (Williams et al., 2021). PPV confidence intervals were also used to generate a range for these 2021 eating disorder prevalence estimates.

3 | RESULTS

3.1 | Study 1: How does the DAWBA eating disorder screen relate to the diagnosis of eating disorders in MHCYP 2017?

3.1.1 | Accuracy of the DAWBA screen assuming no false negatives

Table 1 reports the diagnostic accuracy measures for the standard DAWBA screen threshold for children and their parents in 2017, based

on the assumption of no false negatives. Test accuracy estimates for the standard thresholds were similar for both ICD-10 and DSM-5 diagnoses, with no significant differences (p = .2, t(14) = 1.3). The text below, therefore, refers to DSM-5 diagnoses unless otherwise specified.

Using the standard threshold, the screening questions were highly *specific* in parents: 93.7 (95% CI: 92.8–94.6) for parents of 11–16-year-olds, and 92.7% (95% CI: 89.8–95.1) for parents of 17–19-year-olds, and highly *sensitive* in children and young people at 100%.

Table 2 shows the diagnostic accuracy estimates for parents and their children when a threshold of two positive answers on the screen was applied to both informants. The higher threshold for children and young people increased overall accuracy, but at the expense of increasing the number of false negatives—80% of children/young people who met diagnostic criteria screened positive when a threshold of two or more was applied to both informants. Sensitivity was highest when parent and child answers were combined (Table 2).

3.1.2 | Accuracy of the DAWBA screen assuming a false positive rate of 2.4%

Table 3 depicts the diagnostic accuracy measures for the use of the standard DAWBA screen threshold for children and their parents in 2017 when the false negative rate of 2.4% was applied.

Compared with Table 1, NPV was high for all informants but particularly high for parents, regardless of the false negative rate or threshold. PPV was extremely low for all informants under all conditions, as would be expected given the relatively low prevalence of eating disorders among this age group at population level and the design of the DAWBA to screen with extensive additional assessment of screen positives.

Parental report was highly specific for both age groups and both diagnostic classifications, but sensitivity estimates were imprecise and lower for 11–16-year-olds than 17–19-year-olds. Applying the false negative rate greatly reduced the sensitivity of the screening questions across all informants. Sensitivity was higher when parent and child

TABLE 2 Measures of diagnostic accuracy of the DAWBA score threshold of 2+ applied to children aged 11-16 years, young people aged 17-19 years and combined with their respective parents (where both provided responses) for an eating disorder diagnosis according to DSM-5 and ICD-10 criteria (with 95% confidence intervals) in 2017 based on zero false negatives

	Diagnostic criteria used for clinical	Positive predictive	Negative predictive			
Informant type	rating	value (%)	value (%)	Sensitivity (%)	Specificity (%)	Accuracy (%)
Children aged 11-16 years	DSM-5	1.9 (1.3–2.6)	99.9 (99.7-100.0)	77.8 (40.0-97.2)	85.7 (84.3-87.0)	85.6 (84.2-87.0)
(n = 2597)	ICD-10	1.6 (1.1–2.4)	99.9 (99.7–100.0)	75.0 (34.9-96.8)	85.6 (84.2-87.0)	85.6 (84.2-86.9)
Parents and their children	DSM-5	2.4 (1.9–3.1)	100.0 (99.7–100.0)	88.9 (51.8-99.7)	87.3 (86.0-88.6)	87.3 (86.0-88.6)
11–16 years combined $(n = 2591)$	ICD-10	2.1 (1.6-2.8)	100.0 (99.7-100.0)	87.5 (47.4-99.7)	87.3 (86.0-88.6)	87.3 (86.0-88.6)
Young people aged 17–19 years (n = 935)	DSM-5 & ICD-10 ^a	3.8 (3.3-4.3)	100.0 (-)	100.0 (59.0-100.0)	80.7 (78.0-83.2)	80.8 (78.1-83.3)
Parents and their children aged $17-19$ years combined ($n = 413$)	DSM-5 & ICD-10 ^a	2.5 (2.1-3.1)	100.0 (-)	100.0 (15.8-100.0)	81.3 (77.2-84.9)	81.4 (77.3-85.0)

^aBoth diagnostic criteria detected the same cases.

EATING DISORDERS -WILEY 7

TABLE 3 Measures of diagnostic accuracy of the standard DAWBA score threshold and an eating disorder diagnosis according to DSM-5 and ICD-10 criteria for children aged 11–16 years old, young people aged 17–19 years old and their respective parents (with 95% confidence intervals) in 2017 based on applied false negative rate from clinical sample (2.4%)

	Diagnostic criteria used for clinical	Positive predictive	Negative predictive			
Informant type	rating	value (%)	value (%)	Sensitivity (%)	Specificity (%)	Accuracy (%)
Parents of 11–16-year-	DSM-5	4.9 (1.9-7.9	97.5 (96.8–98.0)	11.8 (4.9–18.6)	93.6 (92.7-94.5)	91.4 (90.3–92.3)
olds (n = 3112)	ICD-10	4.4 (1.6–7.2)	97.4 (96.8–98.0)	10.7 (4.1–17.3)	93.6 (92.7-94.5)	91.3 (90.3–92.3)
Children aged 11–16 years	DSM-5	1.0 (.6–1.8)	97.6 (97.3-97.9)	18.0 (8.6–31.4)	65.1 (63.2-67.0)	64.2 (62.3-66.0)
(n = 2597)	ICD-10	.9 (.5–2.5)	97.6 (97.3-97.9)	16.3 (7.3–29.7)	65.1 (63.2-66.9)	64.2 (62.3-66.0)
Parents and their children	DSM-5	.8 (.4–1.4)	97.6 (97.2-98.0)	20.9 (10.0-36.0)	54.6 (52.6-56.5)	54.0 (52.1-56.0)
11–16 years combined $(n = 2591)$	ICD-10	.7 (.4–1.3)	97.6 (97.2-97.9)	19.1 (8.6-34.1)	54.6 (52.6-56.5)	54.0 (52.1-55.9)
Parents of 17–19-year- olds (n = 415)	DSM-5 & ICD-10 ^a	6.3 (1.8–19.7)	97.7 (96.9–98.2)	18.2 (2.3–51.8)	92.6 (89.6–94.9)	90.6 (87.4-93.2)
Young people aged 17-19 years old (n = 935)	DSM-5 & ICD-10 ^a	1.7 (.9–3.0)	97.7 (96.7-98.3)	36.8 (16.3-61.6)	54.7 (51.4-58.0)	54.3 (51.1-57.6)
Parents and their children aged 17–19 years combined (n = 413)	DSM-5 & ICD-10 ^a	.9 (.3–2.9)	97.5 (96.0-98.4)	28.6 (3.7-71.0)	47.3 (42.4-52.3)	47.0 (42.1-51.9)

^aBoth diagnostic criteria detected the same cases.

TABLE 4 Measures of diagnostic accuracy of the DAWBA score threshold of 2+ applied to children aged 11-16 years, young people aged 17-19 years and combined with their respective parents (where both provided responses) for an eating disorder diagnosis according to DSM-5 and ICD-10 criteria (with 95% confidence intervals) in 2017 based on applied false negative rate from clinical sample (2.4%)

Informant type	Diagnostic criteria used for clinical rating	Positive predictive value (%)	Negative predictive value (%)	Sensitivity (%)	Specificity (%)	Accuracy (%)
Children aged 11–16 years	DSM-5	1.9 (.9–3.7)	97.5 (97.3–97.7)	11.3 (4.7–21.9)	85.4 (83.9-86.7)	83.6 (82.1-85.0)
(n = 2597)	ICD-10	1.6 (.7-3.4)	97.5 (97.3–97.7)	9.8 (3.7–20.2)	85.3 (83.9-86.7)	83.6 (82.1-85.0)
Parents and their children	DSM-5	2.4 (1.3-4.5)	97.6 (97.3–97.8)	12.7 (5.7–23.5)	87.1 (85.7–88.4)	85.3 (83.8-86.6)
11–16 years combined $(n = 2591)$	ICD-10	2.1 (1.0-4.1)	97.6 (97.3-97.8)	11.3 (4.7–21.9)	87.0 (85.7-88.3)	85.2 (83.8-86.6)
Young people aged 17–19 years old (n = 935)	DSM-5 & ICD-10 ^a	3.8 (2.0-6.9)	97.6 (96.9-98.1)	28.0 (12.1-49.4)	80.3 (77.5-82.8)	78.9 (76.1-81.4)
Parents and their children aged 17–19 years combined ($n = 413$)	DSM-5 & ICD-10 ^a	2.5 (.7-8.4)	97.6 (96.8–98.2)	20.0 (2.5–55.6)	80.9 (76.7-84.6)	79.4 (75.2-83.2)

^aBoth diagnostic criteria detected the same cases.

answers were combined, and even higher when the threshold for children was two or more compared to parents as single informants (see Table 4 below). The higher threshold for young people increased overall accuracy, but at the expense of increasing false negatives.

3.2 | Study 2: Estimating prevalence by applying the PPVs to those who screened positive in the 2021 MHCYP follow-up survey

Table 5 displays crude eating disorder prevalence estimates for 11–19-year-olds in 2021 generated by applying our estimated PPVs and 95% confidence intervals to the proportion of children and young people screening positive in the 2021 MHCYP surveys. The overall

estimates show little difference between 2017 and 2021. However, the range of prevalence estimates for young men aged 17-19 was between .4% and 1.2% compared to a prevalence estimate of below .1 in 2017, while for younger boys, the lower limit of the range in 2021 equates to the 2017 prevalence. Similarly, the point prevalence estimate for young women represents a 20% increase in prevalence between 2017 and 2021, but the range of estimates varies from a 30% decrease to nearly doubling.

4 | DISCUSSION

We aimed to provide empirical context to explain the increased clinical presentations of young people with eating disorders through

TABLE 5 Application of estimated PPVs from 2017 to children and young people screening positive in the 2021 MHCYP

	Proportion screening	DSM-5/ICD-10 eating disorder prevalence, 2017 (%)	Proportion screening positive in 2021 MHCYP survey (%)	Estimates of DSM-5/ICD-10 eating disorder prevalence using PPV estimates, 2021 (%)				
N Si	MHCYP survey (%)	Estimate		Estimate	Lower limit (from 95% CI for PPV)	Upper limit (from 95% CI for PPV)		
11–16-year-olds (parent report)								
All	6.5	.6	13.0	.6	.2	1.0		
Girls	8.4	1.0	17.8	.9	.3	1.4		
Boys	5.1	.2	8.4	.4	.2	.7		
17–19-year-olds (self-report)								
All	45.1	.8 ^a	58.2	1.0	.5	1.7		
Girls	60.5	1.6 ^a	76.4	1.3	.7	2.3		
Boys	29.6	.0 ^a	41.0	.7	.4	1.2		

^aPrevalence figures for 2017 draw on a mix of 17–19-year-olds and their parents' reports while in 2021, 17–19-year-olds self-reported only.

exploring the diagnostic accuracy of the DAWBA screen and applying the results to the 2021 reports of eating difficulties. MHCYP needed to apply the same measure in 2017 and 2021 in order to make a direct comparison so we were constrained by the choice of measure.

The DAWBA screening questions were most strong at ruling *out* an eating disorder in children and young people, which is what they were designed to do (Goodman et al., 2000). The DAWBA uses skip rules to balance participant burden against diagnostic accuracy and aims to select out those with no problems in the knowledge that more detailed assessment will support the differentiation of clinical from subclinical disorders. Overall, parent reports were the most diagnostically accurate and specific, which suggests that clinicians can mostly be reassured by a lack of parental concern regarding their child's eating, consistent with the current literature (Ford et al., 2005).

The assessment of eating disorders is complicated as clinical detection relies heavily upon a patient's willingness and ability to share information about their eating behaviors. The denial of symptoms is common, particularly among people with anorexia nervosa (Couturier & Lock, 2006). Although some behaviors are observable, they may not be reported by family, peers and clinicians unless directly enquired about. Furthermore, some of the symptoms required for an eating disorder diagnosis to be made, according to criteria from the DSM-5 or ICD-10, are complex and can be difficult to operationalize in a short screening questionnaire. Existing screening items are either limited to certain age ranges and often exclude younger adolescents to focus on teenagers and adults (e.g., the SCOFF questions [Morgan et al., 1999]), include eight or more items (e.g., Children's Eating Disorder Examination-Questionnaire, ChEDE-Q8 [Kliem et al., 2017]), or focus on particular behaviors (e.g., Adolescent Binge Eating Questionnaire, ADO-BED [Chamay-Weber et al., 2017]). We could benefit from brief general screen that could be used to assess population prevalence as well as for nonspecialists to identify young people who may be struggling with their eating and need further clinical assessment (Zipfel et al., 2022).

Denial and minimization of illness is common among individuals with eating disorders (Starzomska & Tadeusiewicz, 2016; Viglione et al., 2006). Our findings indicate that raising the threshold for young people reduces false positives (fewer unnecessary additional assessments) at the expense of false negatives (more undetected cases). How much this matters depends on the reason for using the screen and opportunities for additional assessment. For example, when positive screens automatically lead to detailed assessment, such as in the 2017 MHCYP survey, minimizing false negatives will ensure more accurate prevalence estimates. In contrast, minimizing false negatives in clinical assessments by school nurses or general practitioners is essential, especially given the low PPV and potential to raise anxiety and swamp clinical services. Eating disorders are highly persistent with a proven mortality and given that their prevalence is probably greatly underestimated, the lower threshold for young people is recommended, with additional assessment by the screening practitioner before referral to specialist services.

Combining parental and child reports increased the sensitivity of the DAWBA screen, but reduced specificity and overall accuracy. Discrepancies between young people and adult informants on mental health symptoms are one of the most robust findings in child and adolescent psychiatry (Collishaw et al., 2009). Information from young people is particularly helpful for detecting concealed or internally experienced difficulties such as self-harm, depression, and anxiety, while information from teachers and parents is more useful with reference to neurodevelopmental and behavioral problems (Aebi et al., 2017; Ford et al., 2005; Kuhn et al., 2017). Both denial of symptoms and lack of insight may also contribute to the lower specificity and moderate sensitivity of the screening questions for young people in our analysis (Keski-Rahkonen et al., 2006; Vandereycken & van Humbeeck, 2008). The strongest predictor of disordered eating behaviors later in adolescence is the degree of disordered eating already present in early adolescence (Attie & Brooks-Gunn, 1989; Wichstrøm, 2000) which suggests that disordered eating, once present, tends not to resolve spontaneously and intensifies over time. Parent or carer reports can therefore still be important and informative during emerging adulthood.

We urge caution when interpreting of the prevalence estimates, which are based on a crude analysis using published proportions rather than raw data. Our 2017 estimates are also inevitably influenced by sampling strategies, age, and the screening tool itself. Nevertheless, we attempted to estimate prevalence for a particular age group in comparison to a particular population-based sample using a particular measure. Both MHCYP surveys report a relatively low prevalence of eating disorders in young people compared to some current literature using other tools (Mitchison et al., 2020; Nagl et al., 2016; Silen et al., 2020). Yet, the prevalence and incidence patterns of eating disorders remain disputed, mainly because of different assessment methods and sampling strategies, as well as recent changes in diagnostic criteria. Swanson et al. reports similar lifetime prevalence of eating disorder to what we found in our study, among boys and girls aged 13-18 using the Composite International Diagnostic Interview (Swanson et al., 2011). In contrast, the German KiGGS-BELLA study used the SCOFF questionnaire to examine disordered eating behavior in a representative sample of 1895 children and adolescents aged 11-17 (Ravens-Sieberer et al., 2008) and found that overall, 29% of the girls and 14% of the boys reported disordered eating behavior indicating that, among adolescents, subthreshold eating disorders might be considerably more frequent compared to threshold eating disorders.

The DAWBA eating disorder module was designed two decades ago and primarily focused on anorexia and bulimia nervosa but was updated to include DSM-5 prior to the 2017 survey. The DAWBA tends to generate fewer, more severe cases when compared to other standardized diagnostic instruments, which is why it is recommended for prevalence surveys (Angold et al., 2012).

Despite only tentative evidence of increased prevalence in eating disorder, the proportion of children and young people who "screened positive" on the DAWBA screening questions rose significantly between 2017 and 2021, which is still worrying (Williams et al., 2021). Children and young people with sub-clinical eating difficulties may still experience impairment and benefit from identification and support. Such symptoms have been strongly associated with other mental health difficulties, including depression, anxiety, substance misuse, and personality disorders (Godart et al., 2007; Hudson et al., 2007; Swanson et al., 2011). It will be important to study the mental health, impairment, and service access trajectory of these children and young people over time to understand better how support and improve their mental health, once the data are available.

Clinical raters could use evidence from other sections of the DAWBA (e.g., body dysmorphia, depression) as well as qualitative data to assign diagnoses if clinical impairment was evident, even if DSM-5 and ICD-10 diagnostic criteria were not met, with only one additional eating disorder emerging in this fashion. Thus, despite the limitations of the DAWBA screen, the very high NPV gives us confidence that those screening negative in the 2017 and 2021 MHCYP surveys are unlikely to have an eating disorder. The 2021 follow-up had to be brief and online, but our work supports the need for inclusion of the full DAWBA in a future baseline population survey with a refreshed and representative sample. Repeating the DAWBA in a larger sample would also provide data on trends in eating disorders, allowing the longer-term impacts of COVID-19 on eating behaviors to be better examined.

9

This is the first study to examine the diagnostic accuracy of the DAWBA screening questions in parents and young people, and the only population-based study to estimate population prevalence of eating disorders before and after the pandemic. It benefits from a large, representative population sample. Nevertheless, the data were not originally collected with the intention of conducting a diagnostic accuracy study, and the present findings must be interpreted with limitations in mind. The full DAWBA was not applied to participants who screened negative in 2017, which deprived Study 1 of a true false-negative rate, necessitating assumptions based on a smaller pilot sample. The "true" false negative rate in the general population of children and young people, however, is likely to fall between our two estimates (0%–2.4%) using this tool.

Further research should analyze the results from the full DAWBA on a large sample of participants who screen negative to ensure more precise estimates. Sensitivity analyses with different assumptions could bolster these results, but the raw data are yet to be made available from NHS Digital.

5 | CONCLUSION

The DAWBA eating disorder screen shows strong NPV and, particularly when used with parents or carers, could be highly useful at ruling out (if negative) and also ruling in (if positive) an eating disorder in clinical practice. While parental report alone provides the highest diagnostic accuracy, our findings also highlight the importance of speaking to both parents and children during assessment and using clinical judgment to balance the evidence when their accounts conflict. Finally, we found tentative evidence of a modest increase in eating disorders at population level.

AUTHOR CONTRIBUTIONS

Tamsin Newlove-Delgado: Resources; writing – original draft; writing – review and editing. Sally McManus: Data curation; project administration. Frances Mathews: Writing – review and editing. Suzanne Hill: Data curation; project administration. Katharine Sadler: Data curation; methodology. Tamsin J Ford: Conceptualization; funding acquisition; supervision; writing – original draft; writing – review and editing.

ACKNOWLEDGMENT

We would like to thank the young people and parents who contributed to the surveys, and Cher Cartwright at NHS digital who coordinated the follow up surveys.

FUNDING INFORMATION

The original survey was commissioned by the Department of Health for England and delivered by a consortium including the National Centre of Social Research (NatCen), Office for National Statistics (ONS), and the universities of Cambridge and Exeter. The UKRI funded the wave 2 follow up and the current analysis: *Tracking the impact of COVID-19 on national longitudinal probability sample MR/V027751/1*. All research at the Department of Psychiatry in the University of Cambridge was supported by the NIHR Cambridge Biomedical Research Centre (BRC-1215-20014) and NIHR Applied Research Centre. Tamsin Newlove Delgado was supported by an NIHR Advanced Fellowship (NIHR 300056). The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care.

CONFLICT OF INTEREST

The authors declare no conflict of interest.

DATA AVAILABILITY STATEMENT

The data that support the findings of this study are available from the UK Data Service via NatCen. Restrictions apply to the availability of these data, which were used under license for this study. Data are available from https://doi.org/10.5255/UKDA-SN-8467-2 with the permission of NHS Digital.

ORCID

10

Jessica O'Logbon ⁽¹⁾ https://orcid.org/0000-0003-1216-9703 Tamsin Newlove-Delgado ⁽¹⁾ https://orcid.org/0000-0002-5192-3724 Sally McManus ⁽¹⁾ https://orcid.org/0000-0003-2711-0819 Frances Mathews ⁽¹⁾ https://orcid.org/0000-0002-1314-4607 Tamsin Ford ⁽¹⁾ https://orcid.org/0000-0001-5295-4904

REFERENCES

- Aebi, M., Kuhn, C., Banaschewski, T., Grimmer, Y., Poustka, L., Steinhausen, H.-C. C., & Goodman, R. (2017). The contribution of parent and youth information to identify mental health disorders or problems in adolescents. *Child and Adolescent Psychiatry and Mental Health*, 11, 23.
- Aebi, M., Kuhn, C., Metzke, C. W., Stringaris, A., Goodman, R., & Steinhausen, H. C. (2012). The use of the development and well-being assessment (DAWBA) in clinical practice: A randomized trial. *European Child and Adolescent Psychiatry*, 21(10), 559–567.
- American Psychiatric Association. (2013). Diagnostic and statistical manual of mental disorders (5th ed., p. 991). American Psychiatric Association.
- Angold, A., Erkanli, A., Copeland, W., Goodman, R., Fisher, P. W., & Costello, E. J. (2012). Psychiatric diagnostic interviews for children and adolescents: A comparative study. *Journal of the American Academy of Child and Adolescent Psychiatry*, 51(5), 506–517. https://doi.org/10. 1016/j.jaac.2012.02.020
- Attie, I., & Brooks-Gunn, J. (1989). Development of eating problems in adolescent girls: A longitudinal study. *Developmental Psychology*, 25(1), 70–79.
- Chamay-Weber, C., Combescure, C., Lanza, L., Carrard, I., & Haller, D. M. (2017). Screening obese adolescents for binge eating disorder in primary care: The adolescent binge eating scale. *Journal of Pediatrics*, 185, 68–72.e1.
- Collishaw, S., Goodman, R., Ford, T., Rabe-Hesketh, S., & Pickles, A. (2009). How far are associations between child, family and community factors and child psychopathology informant-specific and informant-general? *Journal of Child Psychology and Psychiatry*, 50(5), 571–580.
- Couturier, J. L., & Lock, J. (2006). Denial and minimization in adolescents with anorexia nervosa. *International Journal of Eating Disorders*, 39(3), 212–216.
- Feinmann, J. (2021). Eating disorders during the COVID-19 pandemic. BMJ, 374, n1787.
- Fisher, M., Schneider, M., Burns, J., Symons, H., & Mandel, F. S. (2001). Differences between adolescents and young adults at presentation to

an eating disorders program. Journal of Adolescent Health, 28(3), 222–227.

- Ford, T., Sayal, K., Meltzer, H., & Goodman, R. (2005). Parental concerns about their child's emotions and behaviour and referral to specialist services: General population survey. *British Medical Journal*, 331(7530), 1435–1436.
- Godart, N. T., Perdereau, F., Rein, Z., Berthoz, S., Wallier, J., Jeammet, P., & Flament, M. F. (2007). Comorbidity studies of eating disorders and mood disorders. Critical review of the literature. *Journal of Affective Disorders*, 97(1–3), 37–49. https://doi.org/10.1016/ j.jad.2006.06.023
- Goodman, R., Ford, T., Richards, H., Gatward, R., & Meltzer, H. (2000). The development and well-being assessment: Description and initial validation of an integrated assessment of child and adolescent psychopathology. *Journal of Child Psychology and Psychiatry*, 41(5), 645–655.
- Haripersad, Y. V., Kannegiesser-Bailey, M., Morton, K., Skeldon, S., Shipton, N., Edwards, K., Newton, R., Newell, A., Stevenson, P. G., & Martin, A. C. (2021). Outbreak of anorexia nervosa admissions during the COVID-19 pandemic. Archives of Disease in Childhood, 106(3), e15.
- Hudson, J. I., Hiripi, E., Pope, H. G., Jr., & Kessler, R. C. (2007). The prevalence and correlates of eating disorders in the National Comorbidity Survey Replication. *Biological Psychiatry*, *61*(3), 348–358. https://doi. org/10.1016/j.biopsych.2006.03.040
- Keski-Rahkonen, A., Sihvola, E., Raevuori, A., Kaukoranta, J., Bulik, C. M., Hoek, H. W., Rissanen, A., & Kaprio, J. (2006). Reliability of selfreported eating disorders: Optimizing population screening. *International Journal of Eating Disorders*, 39(8), 754–762.
- Kliem, S., Schmidt, R., Vogel, M., Hiemisch, A., Kiess, W., & Hilbert, A. (2017). An 8-item short form of the Eating Disorder Examination-Questionnaire adapted for children (ChEDE-Q8). *International Journal* of Eating Disorders, 50(6), 679–686.
- Kuhn, C., Aebi, M., Jakobsen, H., Banaschewski, T., Poustka, L., Grimmer, Y., Goodman, R., & Steinhausen, H. C. (2017). Effective mental health screening in adolescents: Should we collect data from youth, parents or both? *Child Psychiatry and Human Development*, 48(3), 385–392.
- Meltzer, H., Gatward, R., Goodman, R., & Ford, T. (2003). Mental health of children and adolescents in Great Britain. *International Review of Psychiatry*, 15(1–2), 185–187.
- Mitchell, K. S., Wolf, E. J., Reardon, A. F., & Miller, M. W. (2014). Association of eating disorder symptoms with internalizing and externalizing dimensions of psychopathology among men and women. *International Journal of Eating Disorders*, 47(8), 860–869.
- Mitchison, D., Mond, J., Bussey, K., Griffiths, S., Trompeter, N., Lonergan, A., Pike, K. M., Murray, S. B., & Hay, P. (2020). DSM-5 full syndrome, other specified, and unspecified eating disorders in Australian adolescents: Prevalence and clinical significance. *Psychological Medicine*, 50(6), 981–990. https://doi.org/10.1017/S0033291719000898
- Morgan, J. F., Reid, F., & Lacey, J. H. (1999). The SCOFF questionnaire: Assessment of a new screening tool for eating disorders. *British Medical Journal*, 319(7223), 1467–1468.
- Moya, T., Fleitlich-Bilyk, B., Goodman, R., Nogueira, F. C., Focchi, P. S., Nicoletti, M., Pinzon, V., Cordás, T. A., & Neto, F. L. (2005). The eating disorders section of the development and well-being assessment (DAWBA): Development and validation. *Revista Brasileira de Psiquiatria*, 27(1), 25-31.
- Nagl, M., Jacobi, C., Paul, M., Beesdo-Baum, K., Höfler, M., Lieb, R., & Wittchen, H. U. (2016). Prevalence, incidence, and natural course of anorexia and bulimia nervosa among adolescents and young adults. *European Child & Adolescent Psychiatry*, 25(8), 903–918. https://doi. org/10.1007/s00787-015-0808-z
- Newlove-Delgado, T., McManus, S., Sadler, K., Thandi, S., Vizard, T., Cartwright, C., & Ford, T. (2021). Child mental health in England before and during the COVID-19 lockdown. *The Lancet Psychiatry*, 8(5), 353–354.

11

- NHS England. (2022). Children and young people with an eating disorder waiting times. Retrieved April 1, 2022, from https://www.england.nhs.uk/statistics/statistical-work-areas/cyped-waiting-times/
- Petkova, H., Simic, M., Nicholls, D., Ford, T., Prina, A. M., Stuart, R., Livingstone, N., Kelly, G., MacDonald, G., Eisler, I., Gowers, S., Barrett, B. M., & Byford, S. (2019). Incidence of anorexia nervosa in young people in the UK and Ireland: A national surveillance study. *BMJ Open*, 9(10), e027339.
- Potterton, R., Richards, K., Allen, K., & Schmidt, U. (2020). Eating disorders during emerging adulthood: A systematic scoping review. *Frontiers in Psychology*, 10, 3062.
- Ravens-Sieberer, U., Kurth, B. M., KiGGS study group, & BELLA study group. (2008). The mental health module (BELLA study) within the German Health Interview and Examination Survey of Children and Adolescents (KiGGS): Study design and methods. *European Child & Adolescent Psychiatry*, 17(Suppl 1), 10–21.
- Silen, Y., Sipila, P. N., Raevuori, A., Mustelin, L., Marttunen, M., Kaprio, J., & Keski-Rahkonen, A. (2020). DSM-5 eating disorders among adolescents and young adults in Finland: A public health concern. *The International Journal of Eating Disorders*, 53, 790–801. https://doi.org/10.1002/eat.23236
- Starzomska, M., & Tadeusiewicz, R. (2016). Pitfalls in anorexia nervosa research: The risk of artifacts linked to denial of illness and methods of preventing them. *Psychiatria Danubina*, 28(3), 202–210.
- Swanson, S. A., Crow, S. J., Le Grange, D., Swendsen, J., & Merikangas, K. R. (2011). Prevalence and correlates of eating disorders in adolescents. Results from the national comorbidity survey replication adolescent supplement. Archives of General Psychiatry, 68(7), 714– 723. https://doi.org/10.1001/archgenpsychiatry.2011.22
- Vandereycken, W., & van Humbeeck, I. (2008). Denial and concealment of eating disorders: A retrospective survey. *European Eating Disorders Review*, 16(2), 109–114.
- Viglione, V., Muratori, F., Maestro, S., Brunori, E., & Picchi, L. (2006). Denial of symptoms and psychopathology in adolescent anorexia nervosa. *Psychopathology*, 39(5), 255–260.

- Vizard, T., Sadler, K., Ford, T., Merad, S., Brodie, E., Forbes, N., Pearce, N., Goodman, R., Goodman, A., & Mcmanus, S. (2018). Mental Health of Children and Young People in England, 2017 Survey Design and Methods Report.
- Wichstrøm, L. (2000). Psychological and behavioral factors unpredictive of disordered eating: A prospective study of the general adolescent population in Norway. *International Journal of Eating Disorders*, 28(1), 33–42.
- Williams, T., Robertson, K., Newlove-Delgado, T., Sadler, K., Vizard, T., Cartwright, C., Welsby, M., Cross, L., Mathews, F., Marcheselli, F., McManus, S., & Ford, T. (2021). *Mental Health of Children and Young People in England*, 2021: Survey Design and Methods. NHS Digital.
- World Health Organization. (1992). The ICD-10 classification of mental and behavioural disorders: Clinical descriptions and diagnostic guide-lines. https://www.who.int/publications/i/item/9241544228
- Zipfel, S., Schmidt, U., & Giel, K. E. (2022). The hidden burden of eating disorders during the COVID-19 pandemic. *The Lancet Psychiatry*, 9(1), 9–11.

SUPPORTING INFORMATION

Additional supporting information can be found online in the Supporting Information section at the end of this article.

How to cite this article: O'Logbon, J., Newlove-Delgado, T., McManus, S., Mathews, F., Hill, S., Sadler, K., & Ford, T. (2022). How does the increase in eating difficulties according to the Development and Well-Being Assessment screening items relate to the population prevalence of eating disorders? An analysis of the 2017 Mental Health in Children and Young People survey. *International Journal of Eating Disorders*, 1–11. https://doi.org/10.1002/eat.23833